Author's response to reviews

Title: Detection of Escherichia coli, Salmonella species, and Vibrio cholerae in tap water and bottled drinking water in Isfahan, Iran

Authors:

Hassan Momtaz (hamomtaz@yahoo.com)
Farhad Safarpour Dehkordi (dr.farhads@yahoo.com)
Amin Asgarifar (aminasgarifar@hotmail.com)

Version: 4 Date: 25 February 2013

Author's response to reviews: see over
Answers to the comments:
Dear Reviewers
Good Day
Thanks a lot from your all favors to s. your comments are elaborate and we used from all of them to improve our article. In this belo we listed our answer to you comments. A: means answer. In the mail text the corrections have been highlighted with yellow color.
Thanks a lot again
Prof. Hassan Momtaz
DVM, ph.D., Department of Microbiology, College of Veterinary Medicine, ShahreKord Branch, Islamic Azad University, ShahreKord-Iran, P.O. Box: 166, Tel. Fax: 0098 381 3361064, E. Mail: hamomtaz@yahoo.com, hamomtaz@iaushk.ac.ir

1. WHAT IS THE FLOWING SOURCE OF WATER THE AUTHORS ARE MENTIONING WHEN THEIR SAMPLE IS ONLY BOTTLED DRINKING WATER?

A: as we said in the samples collection the source of water drinking water is a water of river.

2. SALMONELLA ARE VERY UNLIKELY TO DEVELOP WHEN PLACED ON A SELECTIVE MEDIUM AS MOST OF THEM MAY BE IN AN INJURED STATE. THE AUTHORS SAY - THOUGH NOT CLEAR - THAT THEY HAVE PLACED THE FILTERS IN SELENITE F BROTH (NO MORE USED NOW, HOWEVER THIS COULD BE CORRECTED IN THEIR FURTHER RESEARCH), BUT NO MENTIONING ABOUT TIME OR TEMPERATURES OF INCUBATION ON SELENITE F BROTH.

A: the methods which were used for culture techniques have been changed based on your comments. Please see the part “Bacterial isolation and biochemical tests for the identification of E. coli, Salmonella spp., and V. cholera”.

3. AS INDICATED IN MY PREVIOUS REVIEW THIS MANUSCRIPT SUFFERS FROM SERIOUS ERRORS RELATED TO METHODOLOGY. PUBLICATION OF SUCH WRONG METHODOLOGY WILL LEAD TO FURTHER PROPAGATION OF SUCH WRONG METHODS.

A: the methods which were used for culture techniques have been changed based on your comments.
LANGUAGE OF THE MANUSCRIPT ALSO SUFFERS FROM SERIOUS
TYPOGRAPHICAL AND GRAMMATICAL MISTAKES.
A: it has been corrected.

Thanks a lot again.